Author’s response to reviews

Title: Analysis of surgical and oncological Outcome in internal and external Hemipelvectomy in 34 Patients above the Age of 65 Years at a mean Follow-up of 56 Months

Authors:

Wiebke K Guder (wiebke.guder@ukmuenster.de)
Jendrik Hardes (jendrik.hardes@ukmuenster.de)
Georg Gosheger (georg.gosheger@ukmuenster.de)
Marcel-Philipp Henrichs (marcel-philipp.henrichs@ukmuenster.de)
Markus Nottrott (markus.nottrott@ukmuenster.de)
Arne Streitbürger (streitb@ukmuenster.de)

Version: 6 Date: 17 January 2015

Author’s response to reviews:

Dear Ladies and Gentlemen,

we appreciate your comments and have addressed your concerns as follows:

An Acknowledgements Section has been inserted between the Authors' Contribution section and Reference List.

The section Patient Consent (Methods) has been rephrased, addressing your concern of patients consenting to their personal data being published.

Also, in Table 2a the column "gender" was erased as it was not strictly necessary for the interpretation of the data included and by leaving it out we are able to further increase patients’ anonymity.

We are looking forward to your reply. If you have any further concerns please don’t hesitate to contact us.

Sincerely,

W. Guder